Author’s response to reviews

Title: IMPROVDENT: IMPROVING DENTURES FOR PATIENT BENEFIT A crossover randomised clinical trial comparing impression materials for complete dentures.

Authors:

Janine C Gray (j.c.gray@leeds.ac.uk)
Nuria Navarro-Coy (n.navarro-coy@leeds.ac.uk)
Sue H Pavitt (s.pavitt@leeds.ac.uk)
Claire Hulme (c.t.hulme@leeds.ac.uk)
Mary Godfrey (m.godfrey@leeds.ac.uk)
Helen L Craddock (h.l.craddock@leeds.ac.uk)
Paul A Brunton (p.a.brunton@leeds.ac.uk)
Sarah Brown (medsbro@leeds.ac.uk)
Sean Dillon (s.dillon@leeds.ac.uk)
Gillian Dukanovic (g.dukanovic@leeds.ac.uk)
Catherine Fernandez (c.fernandez@leeds.ac.uk)
Jonathan Wright (j.wright@leeds.ac.uk)
Howard Collier (h.a.collier@leeds.ac.uk)
Shirley Swithenbank (shirley6337@hotmail.co.uk)
Carol Lee (t.p.hyde@leeds.ac.uk)
Paul Hyde (t.p.hyde@leeds.ac.uk)

Version: 2 Date: 6 July 2012

Author’s response to reviews:

Deesha Majithia
Executive Editor, BMC Oral Health

6th July 2012

Dear Deesha Majithia

IMPROVDENT: IMPROVING DENTURES FOR PATIENT BENEFIT

Please find attached a revised version of the manuscript for the IMPROVDENT protocol paper submitted to BMC Oral health for consideration. A revised email address for Carol Lee has now been given as requested.

I have also removed the files which contain proof of ethics and funding from the upload site as requested. I had thought that these documents needed to be uploaded onto the system. I apologise for this error.

Please let me know if you need any further information.

I look forward to hearing from you in due course.

With kind regards
Dr Catherine Fernandez
Head of Trial Management
Dental Translational and Clinical Research Unit